**Author's response to reviews**

**Title:** Serum macrophage migration inhibitory factor reflects adrenal function in hypothalamo-pituitary-adrenal axis in septic patients: an observational study

**Authors:**

Takashi Miyauchi (miyauchi-ygc@umin.ac.jp)
Ryosuke Tsuruta (ryosan-ygc@umin.ac.jp)
Motoki Fujita (motoki-ygc@umin.ac.jp)
Tadashi Kaneko (kaneyui-ygc@umin.ac.jp)
Shunji Kasaoka (skasa-ygc@umin.ac.jp)
Tsuyoshi Maekawa (tmaekawa@po.cc.yamaguchi-u.ac.jp)

**Version:** 7  **Date:** 7 October 2009

**Author's response to reviews:** see over
Author's covering letter for initial submission

Title: Serum macrophage migration inhibitory factor is associated with adrenal function in hypothalamic-pituitary-adrenal axis in sepsis: an observational study

Authors:

Version: 1 Date: 24 July 2009

Comments: see over